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Now where were we?

Last meeting together (22.02.10) e S

BPTX publication on DIP and cogging

chaired by Sophie Baron

Monday, 22 February 2010 from 15:00 to 16:00 (Europe/Zurich)
at CERN ( CCC-LPC )

Last headache
(08.05.12 — DeltaT drop)

Manage v

Description Discussion on Massi and Sophie’s proposal for EPTX publication over DIP, synchro modes and cogging for startup.

Last document from Alick (08.03.11)

. e ——— Data Exchange Between the
(UrmiErE L EPEHESH TS Whil Experiments and the LHC -

D EDMS Id: 1134645 v.1 Data Exchange for reporting -
_ Experimental conditions /,/;\«,c .
Alick MACPHERSON v:’/—”
Specification - Operation
2011-03-08

ATLAS LHCb
Wrong Bucket Flag: Beam1 false - false

Wrong Bucket Flag: Beam2 false - false
BPTX: deltaT of IP (B1-B2) -0.02 ns - 0.10 ns 0.07 ns
Luminous size (x,y) in um 101.9,106.4 1.0, 1.0 45.1,53.2 64.0,63.7
Luminous size (z) in mm 62.6 30.0 56.9 58.5
Lumi Centroid (x,y) in um -12.8,17.1 2.0, 2.0 -996.5,156.4 560.0,-50.3
Lumi Centroid (z) in mm -4.8 10.0 3.6 10.9

Last publication on the Cogging Blog (28.06.11)

— Rapid clock phase drift during Fill 1900

Last RF ramplgO‘4.12.14 — lons)

FM_BC1_FREQ A
LHCb Global Timing Monitor BPIMO1 Status Mode: ON FM_BC1_FREQ | 1h/10s | [Hz1
B ceitaT in IFE ins) i

- Colliding 40.073.050,0
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N - 40.078.850,0
40.078.750,0
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22:00:00  03:0f 11:00:00 14:00:00 17:00:00  21:00:00  01:00:00 04.00:00  08:00:00 11:00:00 Last data point: 04-12-2014 16:04:59 UTC
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14:05 - 14:15 status & inventory of the RF/TTC backbone
14:15 - 14:25 Status of the BPTx systems

14:25 — 14:55 Cogging and Fine Phase Tuning

14:55- 15:25 New LHC Timing Vistar

15:25 - 15:40 AoB



